**Author’s response to reviews**

**Title:** Update of incidence and antimicrobial susceptibility trends of Escherichia coli and Klebsiella pneumoniae isolates from Chinese intra-abdominal infection patients

**Authors:**

Hui Zhang (punchzhanghui@163.com)
Qiwen Yang (yangqiwen81@163.com)
Kang Liao (liaokang1971@163.com)
Yuxing Ni (yuxing_ni@126.com)
Yunsong Yu (yvys119@163.com)
Bijie Hu (hubijie@vip.sina.com)
Ziyong Sun (ziyongsun@hotmail.com)
Wenxiang Huang (wenxiang_huang@163.com)
Yong Wang (sdwangyong@126.com)
Anhua Wu (dr_wuanhua@sina.com)
Xianju Feng (Xjfeng62688@163.com)
Yanping Luo (ypluo301@aliyun.com)
Yunzhuo Chu (cyz6630@163.com)
Shulan Chen (chenshulanbm@yeah.net)
Bin Cao (caobin_ben@163.com)
Jianrong Su (youyilab@163.com)
Qiong Duan (duanqiong92@126.com)
Shufang Zhang (haikouiyuan@126.com)
Haifeng Shao (shf521201@163.com)
Haishen Kong (konghs@163.com)
Bingdong Gui (guibingdong@163.com)
Zhidong Hu (huzhidong27@163.com)
Robert Badal (robert_e_badali@yeah.net)
Yingchun Xu (xuyingchunbm@163.com)

**Version:** 3  **Date:** 29 Nov 2017
Author’s response to reviews:

Editor Comments:

1. Please ensure the corresponding author’s email provided in the submission system is identical to the one given in the title page. Currently, Yingchun Xu’s email in the submission system is ‘xuyingchunbm@163.com’ whilst in the title page it is given as ‘xycpumch@139.com’ twice, once in the Corresponding author’s details and again in the list of email addresses for all authors. Please address this issue and ensure one clear email is provided for the corresponding author.
Response: The corresponding author’s E-mails are xuyingchunbm@163.com and xycpumch@139.com, which are both in use. If I can select one for entire usage, I would prefer to use xuyingchunbm@163.com

2. Please ensure you alter the heading titles according to the correct formatting of a Research Article which can be found in this link: https://bmcinfectdis.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article
Response: We corrected it.

3. Please remove the ‘Funding acquisition’ role as part of the Author Contribution section as this does not qualify as an intellectual contribution to the study.
Response: We removed this funding acquisition in the section.

4. Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.
Response: We have added this into the text with all listed author’s agreements.

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
Response: Thanks, we will do it accordingly.